Author's response to reviews

Title: Influence of socio-economic status on habitual physical activity and sedentary behavior in 8- to 11-year old children

Authors:

Clemens Drenowatz (drenowat@msu.edu)
Joey C Eisenmann (jce@msu.edu)
Karin A Pfeiffer (kap@msu.edu)
Greg Welk (gwelk@iastate.edu)
Kate Heelan (heelanka@unk.edu)
Douglas Gentile (dgentile@iastate.edu)
David Walsh (walsh030@umn.edu)

Version: 5 Date: 20 April 2010

Author's response to reviews:

Thank you for accepting this paper for publication.

the only change to the editorial version is that I put a space (empty line) between the discussion section and the heading "conclusions"